Engineering Reports

WILEY

Contextualizing the current state of research on the use of machine learning for student performance prediction: A systematic literature review

Khalid Alalawi^{1,2} | Rukshan Athauda¹ | Raymond Chiong¹

Revised: 28 February 2023

¹School of Information and Physical Sciences, The University of Newcastle, Callaghan, New South Wales, Australia

²College of Computer Science and Engineering, Taibah University, Medina, Saudi Arabia

Correspondence

Raymond Chiong, School of Information and Physical Sciences, The University of Newcastle, Callaghan, NSW 2308, Australia. Email: raymond.chiong@newcastle.edu.au

Abstract

Today, educational institutions produce large amounts of data with the deployment of learning management systems. These large datasets provide an untapped potential to support and enhance decision-making and operations. In recent times, machine learning (ML) has been applied to develop models utilizing this "big" data to assist in decision-making. This study presents a systematic literature review into the application of ML to predict student performance. A total of 162 research articles from January 2010 to October 2022 were critically reviewed and analyzed by applying Kitchenham's systematic literature review approach. Our analysis categorized the literature predicting students' academic performance into two categories: (i) predicting student performance in assessments, courses or programs, and identifying students at-risk of failing their course/program (129 studies); and (ii) predicting student dropout or retention in a course or program (33 studies). Classification is the most commonly used approach for predicting student performance (138 studies), followed by regression (25 studies) and clustering (9 studies). Supervised learning methods are used more often than semi-supervised learning. Five most popular ML algorithms include the Decision Tree, Random Forest, Naïve Bayes, Artificial Neural Network, and Support Vector Machine. Historical records of students' grades and class performance, academic related data from learning management systems, and students' demographics are the most common features used for predicting students' performance. The most common methods used for feature selection are Information Gain-based selection algorithms, Correlation-based feature selection, and Gain Ratio. The general platforms/tools/libraries used in the studies include WEKA, Python, R, Rapid Miner, and MATLAB. We also investigated possible actions considered in the literature to help at-risk students. We only found very few studies that deployed remedial actions and evaluated their impact on students' performance. In conclusion, ML has shown great

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

© 2023 The Authors. Engineering Reports published by John Wiley & Sons Ltd.

potential in the prediction of student performance, but also has many areas of further research.

K E Y W O R D S

classification, clustering, educational data mining, machine learning, prediction, regression, student performance, systematic review

1 | INTRODUCTION

Today, large amounts of data are produced by educational institutions with the widespread use of online educational systems and learning technologies in the educational domain.¹ Consequently, the potential to extract knowledge and insights from educational data has gained momentum. These insights have the potential to significantly impact students' learning experiences, learning outcomes, efficiency in managing learning organizations' services, and others. Educational Data Mining (EDM) is an emerging field aimed to address these needs.²

EDM is an interdisciplinary field of study that applies Data Mining (DM) methods to educational data in order to solve or gain insights from the massive amount of data in educational systems.³ The International Educational Data Mining Society defines EDM as an evolving field that aims to establish methods to investigate the particular types of data from educational settings and to use these methods to better understand learners and the environments they study in.⁴ DM methods use techniques from statistics, databases, artificial intelligence and, more recently, Machine Learning (ML) to analyses data. While conventional data mining techniques typically require manual intervention, ML does not require human intervention and can continuously learn and improve their models as more data becomes available.

More specifically, ML is a branch of artificial intelligence, which enables systems to automatically learn and optimize models from analyzing data without being directly programmed.⁵ ML focuses on building models that can learn complicated relationships or patterns through accessing large datasets.^{6,7} The models can then be used for prediction. EDM has successfully used ML methods to build models to meet various educational goals.⁸ ML has been successfully applied to predict students' academic performance, identify at-risk students, define key learning requirements for different students, predict student graduation rates, and others.⁹

Predicting student performance is an important area of research in EDM and has been studied extensively. There are a number of papers that have surveyed DM methods on predicting students' performance.¹⁰⁻¹⁴ In the work of Roy and Garg,¹⁰ an analysis of the different DM techniques, pros and cons of DM methods, and DM tools and trends in the EDM literature are discussed. Another study detailing DM techniques for predicting student performance is presented by Christy and Rama.¹¹ In the study by Shahiri and Husain,¹² a systematic literature review (covering 30 studies from 2002 to 2015) on predicting students' performance using DM techniques is discussed. They investigated significant attributes used for prediction, DM techniques, and the accuracy of predicting student performance. A similar study is presented by Kumar et al.¹³ where they presented a systematic literature review covering papers published between 2007 and 2016. Abu Saa et al.¹⁴ also conducted a systematic literature review analyzing 36 studies from 2009 to 2018, with the aim to identify the common factors that influence the prediction of student performance and DM techniques that identify those factors. Albreiki et al.¹⁵ presented a systematic literature review of using EDM to identify student dropouts and students at risk of failure from 2009 to 2021. Their review included 78 research articles. Hellas et al.¹⁶ presented a comprehensive systematic literature review analyzing atticles from 2010 to 2018 to define what performance means and the purpose of predicting students' performance.

Our study complements existing work by conducting an extensive systematic literature review on the use of ML methods to predict student performance in the context of EDM, covering published work from January 2010 to October 2022. Although ML methods have shown promise in predicting student performance, previous literature reviews have considered DM techniques in general. Our study is unique in that we focus on ML methods and algorithms, as well as their context and use in predicting student performance. In addition, we identified features/factors affecting the prediction models and feature selection methods/algorithms applied in the literature. Finally, we have looked at what remedial actions were taken in the literature for those predicted at-risk students. The research questions (RQs) addressed in this study are as follows:

[•] What are the goals/objectives of applying ML techniques in predicting student performance?

- What are the most common ML methods and algorithms applied in the literature for predicting student performance?
- What are the most commonly used features/factors for the prediction of student performance?
- What methods for feature selection are employed in the literature?
- What learning environments have ML methods been applied to predict student performance?
- What are the ML tools/platforms/libraries used to implement ML algorithms for predicting student performance?
- What actions are taken, based on the prediction results, for the identified at risk students?

The above seven RQs help us contextualize the current state of research in using ML to predict students' performance. To the best of our knowledge, there are no other systematic literature reviews that have considered feature selection methods/algorithms applied for predicting students' performance, as well as reviewed studies that take actions based on student performance prediction results.

Through this literature review, we are able to identify the emerging areas and current applications of ML algorithms in predicting students' academic performance, students at-risk of failure or dropping out, and student retention. This review not only highlights the main areas of focus in the literature, but also exposes areas that require further attention. Additionally, the review provides an analysis of the most prominent ML techniques, algorithms, platforms and tools used for predicting student performance.

The rest of this article is organized as follows: Section 2 presents some background on EDM and ML methods. Section 3 details the literature search procedure and criteria. Section 4 analyses the literature, with the goal of addressing the seven RQs outlined above, and reports results. Discussion is presented in Section 5. Section 6 concludes the article with a view on future research directions.

2 | BACKGROUND

EDM is an interdisciplinary field of study that seeks to better understand students' interests and behaviors by applying DM techniques and algorithms to explore data from educational settings.¹⁷ ML methods, a subset of DM techniques popularly used, can be divided into three types: *supervised learning, unsupervised learning*, and *semi-supervised learning*. Supervised learning utilizes input data with labelled responses aiming to develop models that inference relationships between input and output observables in the data.¹⁸ Classification and regression are two main approaches used in supervised learning. Classification has a categorical output label while regression has a numerical value as the output label. In contrast, unsupervised learning utilizes input data without labelled responses in order to find unseen patterns or clusters in the data.^{19,20} Cluster analysis is the most common method used in unsupervised learning.¹⁸ Semi-supervised learning exploits a small pool of labelled data with a large pool of unlabelled data in order to build efficient learning models by extensively analyzing the unlabelled data.²¹ It is typically used when labelled data is small or is not enough to create an efficient model. This section will briefly introduce some popular ML methods used in EDM.

2.1 | Supervised ML methods for classification

Classification is a supervised learning approach that learns from input data, and then uses it to map or classify new observations; the goal is to predict output categorical variables using input data. When learning a model, classification uses labelled input data as a training dataset to build the model. Unseen data is then used to test the model's efficiency and analyze its results by evaluating the accuracy of the model.²² The most common metrics used to evaluate a classification model include the Confusion Matrix, Receiver Operator Characteristic (ROC), and F-measure (F1) Score.²³ Classification is the most popular approach used in dealing with educational data.²² Some of the common ML algorithms used in classification problems are: C4.5, ID3, K-Nearest Neighbour (K-NN), Naïve Bayes (NB), Support Vector Machine (SVM), and Artificial Neural Networks (ANN).

2.2 | Supervised ML methods for regression

Regression is a supervised learning approach used mainly when there is a need to predict a continuous outcome variable (y), which is the dependent variable, based on the value of one or multiple predictor variables (x_i) , called independent

variables, by evaluating the relationship between variables.¹⁰ The goal of regression models is to find an equation that explains (y) as a function of the (x_i) variables, which can then be used to predict the (y) value based on (x_i) variables. The Root Mean Squared Error (RMSE), Mean Absolute Error (MAE) and Adjusted R-square are commonly used to evaluate the efficiency of a regression model. Common ML algorithms used in regression problems in the education domain include Multiple Linear Regression, Polynomial Regression, Decision Tree Regression, Random Forest Regression, and Support Vector Regression.

2.3 Unsupervised ML methods for clustering

Clustering is an unsupervised learning method that divides objects into groups, where objects in the same group are more similar compared to the objects in other groups.¹⁰ The goal of cluster analysis is to gain insights from grouping unlabelled data. Clustering is an unsupervised learning approach mainly because it utilizes unlabelled data.²⁴ K-mean clustering is a widely used algorithm in clustering problems in the education domain. The Silhouette Score and Silhouette Plot are most commonly used metrics for measuring the performance of a clustering algorithm. They are used to measure the separation distance between clusters.

2.4 | Semi-supervised ML methods

Semi-supervised ML algorithms fall between supervised and unsupervised learning. Generally, semi–supervised learning uses a small amount of labelled data and a significant amount of unlabelled data for training, in order to uncover hidden information in the unlabelled data and build effective learning models by analyzing the unlabelled data extensively.²¹ The aim of semi-supervised learning is to consider how the mixture of labelled and unlabelled data will affect the learning behavior and develop algorithms that benefit from such a combination. Semi-supervised learning is important in ML and DM, as it can use accessible unlabelled data to enhance supervised learning activities when labelled data is unavailable or expensive to obtain.^{25,26} Self-training, co-training and tri-training are common semi-supervised techniques.²⁷ Metrics such as the Confusion Matrix, ROC, Area Under the ROC Curve (AUC), and F1 Score²³ are widely used to evaluate the performance of a semi-supervised learning method.

3 | LITERATURE SEARCH PROCEDURE AND CRITERIA

The main objective of this article is to present a systematic literature review on utilizing ML techniques to predict student performance. This study followed Kitchenham et al.'s²⁸ methodological guidelines for conducting a systematic literature review. Kitchenham et al.'s²⁸ methodology presents a comprehensive set of guidelines for conducting systematic literature reviews suited for software engineering researchers. It offers a relatively high-level overview covering three phases of a systematic literature review: planning, conducting, and reporting the review. Kitchenham et al.'s²⁸ methodological guidelines have several advantages, including:

- Rigor and consistency: The guidelines provide a structured and systematic approach to conducting a literature review, which ensures rigor and consistency in the review process.
- Increased transparency: The guidelines provide a clear and transparent method for documenting the literature review process, which makes it easier to evaluate the quality of the review and the conclusions drawn from it.
- Improved reproducibility: By following a structured and systematic approach, the guidelines help to ensure that the literature review can be easily reproduced by others, thereby promoting the transparency and replicability of the research.
- Better evidence synthesis: The guidelines promote the systematic synthesis of evidence from multiple sources, which can lead to more comprehensive and robust conclusions.
- Enhanced reliability: The guidelines help to ensure that the literature review is reliable and free from bias, which increases the credibility and validity of the research findings.

Overall, Kitchenham et al.'s²⁸ methodological guidelines provide a useful methodology for conducting a systematic literature review and can help to improve the quality and rigor of the research. Therefore, we followed Kitchenham's methodological guidelines to conduct this review. In this section, we present the literature search procedure used in detail.

We defined the scope of "student academic performance" broadly, for the purposes of this literature review, to include students' performance on a single assessment, performance in a course, and retention in a program or continued progress through a program/course. However, we only looked at quantifiable variables directly connected to the course or program, such as grades, pass/fail rates, program retention and course/program dropout. We did not include papers that predict factors that are not measurable or directly connected to academic performance, such as depression or team cohesiveness, even if they are likely to impact student performance. We broadly defined "student dropout" as a student who withdraws a course, school, college, or university before completing their qualification. We also broadly defined "student retention" as the percentage of students who re-enroll in courses in a program in the subsequent year. Finally, we broadly considered "early prediction" as predictions that occur before or during the term where predicted low-performing students still have the opportunity to reflect/improve their performance to successfully complete a course or program.

Several keywords to identify papers related to the RQs were used. Specifically, the terms "educational data mining," "academic performance," "student performance," and "machine learning" were utilized in Scopus, Web of Science, and the ACM Digital Library to search for relevant studies. Boolean operators such as AND, OR and NOT were used in the search strings. This study covered papers published from January 2010 to October 2022. The inclusion criteria used to examine relevant studies are listed below:

- 1. Papers that have used ML methods and algorithms in the context of education settings.
- 2. Papers that are either journal or conference publications.
- 3. Included papers must not be a review paper.
- 4. Papers published in English.

The exclusion criteria used to exclude studies that are not relevant are as follows:

- 1. Papers that are not relevant to the RQs.
- 2. Papers that do not use ML methods and algorithms.

The approach of selecting studies that are relevant to the RQs had three steps: First, the selection of published papers that satisfy the search keywords or search strings; the search query was: ("educational data mining" AND ("academic performance" OR "student performance") AND "machine learning"). Then, a selection of papers based on the selection criteria by reading the title, abstract, and keywords of each paper. The final process was to select papers based on reading their full text.

The process of selecting relevant studies was performed by the first author. The second and third authors checked and confirmed the results to ensure and reduce bias from the first author. The final selection included 162 papers. Figure 1 explains the process of selecting the literature in this review. A summary of the reviewed papers in terms of their aims of prediction, ML methods, datasets/data sources, types of features, feature selection methods, and platforms/tools used to implement the ML algorithms is provided as supplementary material to this paper. In addition, we have listed the top 25 cited research articles in Table 1, and a year wise distribution of included publications in Figure 2.

4 | RESULTS

In this section, we report the results of our systematic literature review study, where we address our RQs and elaborate on the findings derived from analyzing the extracted data.

4.1 | RQ 1: ML and students' performance

Analyzing the results of the systematic literature review, we can broadly classify the main objectives of applying ML to predict student performance into two categories:



FIGURE 1 An overview of the literature selection process.

- Predicting students' academic performance in a course or program and identifying students at risk of failing.
- Predicting students' academic performance in a course or program and identifying students at risk of dropping out.

Predicting students' academic performance includes applying ML algorithms to predict marks/grades in assessments and courses, predict the Cumulative Grade Point Average (CGPA), predict students' graduation time, and identify students who are at risk of failing their courses/programs. There were 129 studies in this category (see Figure 3). Identifying students at risk of failing early on can assist in intervention strategies aimed to achieve better outcomes. A second category of applying ML algorithms to predict student performance is to predict whether a student will drop out of a course, a program, or a university. There were 33 studies classified in this category (see Figure 3).

4.1.1 | Predicting students' academic performance and detecting students at risk of failing

This category has attracted the attention of many researchers, to understand students' learning processes and behaviors with the aim to improve students' performance and outcomes. Various researchers used ML algorithms, both supervised and unsupervised learning, to extract hidden knowledge and patterns from student data.⁵³ Many applications used classification, regression, or clustering methods in educational datasets to predict students' performance in their exams, predict if a student will be able to pass certain courses or not, predict students' CGPA, predict students' graduation time, and others. A high-level analysis of some studies that predict students' performance is discussed below.

Predicting students' performance in their assessments is crucial for academic institutions due to the effective actions that could be taken by identifying students at risk of failing in the early stages and providing interventions early on

ngineering Reports

TABLE 1 Top 25 most-cited articles in this literature review.

#	References	Title	No. of citations ^a
1.	29	A combinational incremental ensemble of classifiers as a technique for predicting students' performance in distance education	131
2.	30	Use of machine learning techniques for educational proposes: A decision support system for forecasting students' grades	123
3.	31	Recommender system for predicting student performance	117
4.	32	Exploring machine learning methods to automatically identify students in need of assistance	114
5.	33	Using machine learning to predict student difficulties from learning session data	93
6.	34	Predicting students' performance in educational data mining	91
7.	35	The impact of engineering students' performance in the first three years on their graduation result using educational data mining	72
8.	36	Application of machine learning in predicting performance for computer engineering students: A case study	63
9.	37	Deep knowledge tracing on programming exercises	61
10.	38	Comparing different resampling methods in predicting students' performance using machine learning techniques	54
11.	39	Data mining approach to predicting the performance of first year student in a university using the admission requirements	52
12.	40	Can online discussion participation predict group project performance? Investigating the roles of linguistic features and participation patterns	52
13.	41	Systematic ensemble model selection approach for educational data mining	50
14.	42	Learning to represent student knowledge on programming exercises using deep learning	49
15.	43	Transfer learning from deep neural networks for predicting student performance	47
16.	1	Student academic performance prediction using supervised learning techniques	46
17.	44	Analysis of educational data mining using classification	42
18.	45	Predicting secondary school students' performance utilizing a semi-supervised learning approach	35
19.	46	EMT: Ensemble meta-based tree model for predicting student performance	34
20.	47	Tracking student performance in introductory programming by means of machine learning	33
21.	48	Interpretable multiview early warning system adapted to underrepresented student populations	32
22.	49	Ensemble learning for estimating individualized treatment effects in student success studies	31
23.	50	Implementing AutoML in educational data mining for prediction tasks	29
24.	51	Multi-split optimized bagging ensemble model selection for multi-class educational data mining	28
25.	52	Student performance prediction model based on supervised machine learning algorithms	28

^a No. of citations was retrieved from Scopus and Web of Science on 10/2/2023.

to assist in improving their performance.⁵⁴ Extensive research has been conducted to predict students' performance in assessments. For instance, Kostopoulos, et al.⁵⁵ carried out a study to predict undergraduate students' performance (either pass or fail) in the final examination at the end of the academic year in an e-learning course using a co-training method based on two views of the dataset. The first view includes features concerning students' demographic and academic performance; and the second view includes features on tracking students' online activities in the course's learning management system (LMS). The researchers used five ML algorithms (K-NN, Extra Tree Classifier, Random Forest, Gradient Boosting Classifier (GBC), and Gaussian NB Classifier) as base learners to form a number of co-training and self-training algorithms. The results showed that co-training outperforms self-training algorithms. More specifically, their proposed co-training (Extra, GBC) algorithm provided the highest accuracy (77.69%–85.19%, confirmed by the Friedman Aligned Ranks Nonparametric test, and F1 scores ranged from 0.767 to 0.847).



FIGURE 2 Year wise distribution of included publications.



FIGURE 3 Studies grouped by their purpose of prediction (percentage based on all reviewed studies). Percentages: number of studies in each category divided by the total number of studies (which is 162), and then multiplied by 100.

In a similar study, Athani et al.⁵⁶ classified students' final exam results into five levels ("A," "B," "C," "D," and "F") using two ML algorithms: Multiclass SVM (MSVM) and ANN. The results showed that the MSVM outperforms ANNs (accuracy of 84.16% for the MSVM and 68.9% for ANNs). Another study was conducted by Kotsiantis et al.²⁹ to predict students' performance in a distance learning environment by implementing an ensemble classification method that combines an incremental version of NB, the WINNOW algorithms using the voting methodology, and 1-NN. Ensemble methods are techniques that seek to improve the performance of a model by training multiple models and combining their predictions, instead of relying solely on a single model. The combined models generally improve the results' accuracy of a model.^{57,58} The dataset used in this study consisted of 1347 instances and five features (1st written assignment, 2nd written assignment, 4th written assignment, and the final exam grades) from an online Informatics course. Results showed that the proposed ensemble method outperforms the NB, 1-NN and WINNOW algorithms as well as other state-of-art ML algorithms (Back-Propagation, 3-NN, Radial Basis Function Algorithm, C4.5, Sequential Minimal Optimization, and RIPPER). The average accuracy of the proposed ensemble method was 78.95%.

EDM can play a significant role in finding issues that affect students' performance, especially those related to students who are at risk of failing. A number of studies have presented early warning systems or models to identify students at risk of failing. The first step towards improving academic performance is by early detection of low-performing students in the academic term. Early warning systems are considered to be effective tools for early detection of students who are at risk.⁵⁹ Through the use of the warning systems, educators can deliver early intervention strategies and real-time support to assist struggling students as well as adjust the course to cater to students' needs.⁶⁰

Many studies applied ML techniques on educational data to predict at-risk students who might fail their courses or programs. For instance, Trakunphutthirak et al.⁶¹ used a university Internet access log file to predict students' academic performance and detect students who are at risk of failing from their browsing behavior. The study applied five ML algorithms: Decision Tree, NB, Logistic Regression, ANN, and Random Forest. The average accuracy of Decision Tree was 76.7%, NB was 58.3%, Logistic Regression was 68.9%, ANN was 73.9%, and Random Forest was 78.9%. The Random Forest technique provided the highest accuracy to predict students at risk of failure. Another study conducted by Iqbal et al.⁶² proposed an early prediction of students' grades in order to provide special attention for at-risk students at the early stages in the semester. The study evaluated multiple ML techniques for predicting students' grades, such as Matrix Factorization, Nuclear Norm Minimisation, Collaborative Filtering, Bayesian Ridge Regression, and the Restricted Boltzmann Machine (RBM). Results indicated that the RBM provides best results in terms of RMSE and MAE for predicting students' grades.

In another study, Cano and Leonard⁴⁸ presented a multi-view early warning system that aims to identify students who are at risk of failure, course withdrawal and dropping out. The study built an incremental model on a weekly basis using Multi-View Genetic Programming (MVGP) classification. The study evaluated their model's performance against single-view algorithms. The average accuracy of single-view algorithms ranged between 82.5% and 91.1%, while the average accuracy of their proposed MVGP was 91.7%. In addition, the average AUC values of the single-view algorithms were between 70.4% and 86.3%, while the average AUC of the proposed MVGP was 90.1%.

We observe that most work along this line of research used classification methods over regression for classifying students' results at the end of their courses. This makes sense since the nature of the problems is mostly to categorize students into different levels such as pass or fail as in binary classification or low, average and high performance in multiclass classification. Moreover, many different ML algorithms have been utilized for the case of predicting student academic performance. Various factors affect the choice of ML algorithms, such as the type of problem, size, quality, nature of data, number of features or parameters, and linearity.

4.1.2 | Predicting students' retention and dropout

Another critical issue for academic institutions is student retention (in courses, programs, etc.) and successful completion (e.g., graduating on time). Low student retention rates can have significant implications on tertiary institutions. According to the Complete College America dashboard,* only 5% of full-time students at 2-year institutions graduate on time, 12% of full-time students at 1-to 2-year certificate programs graduate on time, and only 20% of full-time American college students in 4-year Bachelor degree programs graduate on time.

Several studies have applied ML techniques to predict students' graduation time, to identify students who are in danger of not graduating from their programs within the expected time. The work of Livieris et al.⁶³ proposed a semi-supervised self-trained two-level classification algorithm to predict whether a student will graduate within 6 years or not; followed by predicting the student's graduation time if they will graduate within 6 years. The study applied several ML algorithms on their proposed model: NB, Multi-Layer Perceptron, Sequential Minimal Optimization, K-NN, C4.5, and RIPPER (JRip). Results indicated that the proposed model can identify students at risk of not graduating and accurately classify students based on their graduation time in either four years, five years, or six years. More specifically, K-NN showed the best average classification accuracy results of 66.17%, 77.92%, and 81.48% utilizing 20%, 30%, and 40% labelled data ratios, respectively.

Student dropout is also a critical issue faced by higher education institutions in many countries. Reducing the student dropout rate has been a common goal of many universities. A number of studies have applied ML techniques to predict whether or not a student will dropout from a course. In the work of Burgos et al.,⁶⁴ a study was conducted to predict whether a student will drop out of a course in e-learning or not using a dataset of 104 students in five courses. The study built an incremental model using Logistic Regression to detect if a student is at risk of dropping out of a course. The study also created a tutoring plan to advise students who are at risk of dropping out. The proposed model predicts students' performance based on their activities' grades every week. The model was able to achieve an accuracy of 97.13%, precision of 98.95%, recall of 96.73%, and specificity of 97.14% at week 10 (middle of the term). The authors were able to reduce the

ALALAWI ET AL

dropout rate by 14% compared with previous academic years by implementing dropout prevention actions at different times during the term targeting identified at-risk students.

Another study conducted by Iam-On and Boongoen⁶⁵ presented a new data transformation model, based on the summarized data matrix of link-based cluster ensembles, to improve the classification accuracy of predicting students' dropout via converting the original data to a new form. The study used a dataset with 811 records and 21 features from an operational database system at Mae Fah Luang University, Thailand, from 2009 to 2012. The aim was to predict students' dropout before and after the first year at the university. Their results indicated that the proposed method has an accuracy level of 91% and performed better than many well-known reduction methods.

In summary, both student retention and dropout are of significant concern in academic institutions due to their impact on student graduation. About 30% of first-year students at the United States' baccalaureate institutions do not return for their second year.⁶⁶ As discussed, several studies have used ML algorithms to identify students who are at risk of dropping out. A dropout early warning system can assist universities in identifying and responding to students who are at risk of dropping out of their courses or university programs. At-risk students tend to drop out without proper evaluation of the negative implications of their decisions or without having the chance to seek expert advice and support. The dropout early warning systems can guide students who are likely to drop out with intervention strategies that can lead them to successful completion and graduation.⁶⁷

4.1.3 | Summary

The above analysis addressed our first RQ—"What are the goals/objectives of applying ML techniques in predicting student performance?" We can classify broadly the main goals/aims of applying ML algorithms and methods for predicting student performance (in assessments, courses, or programs) as: (i) predicting student academic performance and identifying students at risk of failing, and (ii) predicting students at risk of dropping out. What is also interesting to note is that most of the research studies have focused on prediction of academic performance for detecting students at risk of failing (129 studies), while fewer studies have focused predicting students at risk of dropping out (33 studies). These findings thus highlight areas where future research can devote attention to, such as student dropout and retention, given that it is a prevalent and significant issue.⁶⁶

The next section looks at the main ML methods/algorithms used for predicting student performance.

4.2 | RQ 2: ML methods and algorithms

There are three main approaches utilized for predicting students' performance in the literature: classification, regression, and clustering. Classification is the main approach used in the literature to predict student performance (around 80% of studies, see Figure 4). In our survey, 25 studies used regression to predict student performance. Only nine studies applied



clustering (i.e., unsupervised) for predicting student performance. Our analysis (as detailed in Figure 4) indicates that most research work examined supervised learning as classification or regression methods, while only six studies utilized semi-supervised learning in classification and two studies used regression. Note: some studies used more than one method and thus are classified into multiple groups.

We have extracted more than 50 distinct ML algorithms used by the 162 papers reviewed. Nonetheless, the most widely used algorithms for predicting students' performance that were applied in ten or more studies are shown in Figure 5. As can be seen in Figure 5, the Decision Tree, Random Forest, NB, ANN, and SVM are the top five mostly used ML algorithms in the literature. Based on this analysis, we are able to answer the second RQ—"*What are the most common ML methods and algorithms applied for predicting student performance in the literature?*"

ML algorithms' computation time complexity plays a vital role in selecting an ML algorithm to create prediction models. Each ML algorithm takes time to train a set of data and perform a prediction, which refers to the computational complexity of ML algorithms. Table 2 shows the time complexity of different ML algorithms widely referred to in this review. We noted that NB and Decision Trees are the least complex algorithms in Table 2. This could explain why these two algorithms are also the most used algorithms.



FIGURE 5 Frequency of the most common ML algorithms used in the literature.

TABLE 2 Computation complexity of ML algorithms.

Algorithm	Classification/ Regression	Training	Prediction
Decision Tree	C + R	O(n ² p)	O(p)
Naive Bayes	С	O(np)	O(p)
Random Forest	C + R	$O(n^2 p n_{trees})$	O(pn _{trees})
Gradient Boosting (n _{trees})	C + R	O(npn _{trees})	O(pn _{trees})
Linear Regression	R	$O(p^2n + p^3)$	O(p)
SVM (Kernel)	C + R	$O(n^2p + n^3)$	O(n _{sv} p)
k-Nearest Neighbours (naive)	C + R	O(knd)	O(np)
Neural Network	C + R	-	$O(pn_{l1} + n_{l1}n_{l2} + \dots)$

Note: https://www.thekerneltrip.com/machine/learning/computational-complexity-learning-algorithms/.

Abbreviations: (k), number of nearest neighbours and (d) refers to dimensions; (n), the number of training samples; (p), the number of features; (n_{trees}), the number of trees (for methods based on various trees); (n_{sv}), the number of support vectors; (n_{li}), the number of neurons at layer (i) in a neural network.

4.3 | RQ 3: Common features/factors used for predicting students' performance

We classified each study into one or more categories as described in Table 3. These include: (1) Historical records of students' grades and class performance, (2) Students demographics related features, (3) Academic related features from LMSs, (4) Learning/eLearning activity related features (Log File), (5) Instructor related features, (6) Students' social information, (7) Internet access log files, (8) Parents' related features, (9) Students' economic related features, and (10) Students' behavioral related features. Table 3 describes each of these categories.

As can be seen in Figure 6, the most commonly used features for creating ML models for predicting students' performance are "Historical records of students' grades and class performance" (presented in 122 studies, 27%), followed by "Academic related features from LMSs" (presented in 98 studies, 21.6%), "Students' demographics" (presented in 97 studies, 21.4%), and "Students' social information" (presented in 44 studies, 10%), "Learning/eLearning activity related features" (presented in 37 studies, 8%), and "Students' behavioral information" (presented in 31 studies, 7%). The least two categories—"Internet access log files" (presented in 3 studies, 0.7%) and "Instructor related features" (presented in 2 studies, 0.4%), which represent a total of 3% altogether, were used by a few number of studies; thus, they were considered as ad hoc features.

These findings, especially the top four categories, are in agreement with the findings of a previous systematic literature review conducted by Abu Saa et al.¹⁴ which showed that the historical records of students' grades and class performance, students' eLearning activity, students' demographics, and students' social information are the most common features used for predicting students' performance. Shahiri and Husain¹² systematic literature review results also showed that CGPA and historical records of students' grades and class performance are the most commonly used features, followed by students' demographic features, extra-curricular activities, and students' social related features. Hellas et al.¹⁶ also indicated that course performance features, pre-course performance features, demographic features, and student engagement are the top factors used for predicting students' performance.

#	Category	Description
1	Historical records of students' grades and class performance	Historical records of students' grades on assessments/courses or other performance indicators from previous courses, semesters, or years.
2	Students demographics related features	Student demographic data, including gender, age, nationality, ethnicity, etc.
3	Academic related features from LMSs	Information related to students/courses/programs/universities that are acquired from an LMS, such as the course name, course information, school subjects, major, absence, admission entry requirement, etc.
4	Learning/e-learning activity related features (Log File)	The activity logs of students in eLearning systems, such as the number of logins, time spent on activities, idle time, time spent on students' activities on e-learning system (e.g., forum, etc).
5	Instructor related features	Information related to the instructor of a student and their evaluation results.
6	Students' social information	Information related to students' social life, like the number of friends, free time after school, if s/he smokes or not, etc.
7	Internet access log files	Information related to students' log files of using the Internet, like browsing data, source addresses, session ID, start time, elapsed time, etc.
8	Parents' related features	Information related to students' parents such as parent status, parents' jobs, parents' school satisfaction, parents' cohabitation status, parents' answering survey, etc.
9	Students' economic related features	Information related to students' economic status such as students' income, socio-economic status.
10	Students' behavioral related features	Information related to students' behavioral status such as study time, usage of the Internet TV, or game, students' attention, if a student visited resources, raised hand on class, etc.

TABLE 3 Description of features/attributes' categories.



FIGURE 6 Distribution of studies by feature' categories.

4.4 | RQ4: Feature selection methods

Feature selection is a process that consists of selecting subsets of variables that are relevant to predictive models. It reduces the number of predictor variables used in ML algorithms. Also, it reduces the amount of time it takes to compute ML models, improves their generalization ability, prevents overtraining, and reduces the amount of resources (memory and CPU time) required to predict. A feature selection algorithm entails finding combinations of features that have the greatest probability of increasing the predictive model's accuracy.^{33,68–70} Three main approaches used for feature selection: Filter methods, Wrapper methods, and Embedded methods.⁷¹

- Filter methods employ univariate statistics to determine the intrinsic properties of the features instead of relying on cross-validation measures to assess their performance. There is a considerable advantage to these methods in terms of speed and computational load compared to wrapper methods. When dealing with high-dimensional data, the use of filter methods is more cost-effective from a computational standpoint.^{68–70,72} Information Gain, Chi-square Test, Fisher's Score, and Correlation Coefficient are the most often used filter methods.
- Wrappers require some method of searching through the space of all possible subsets of features and assessing their quality by learning and evaluating a classifier that can be applied to that subset of features. ML algorithms are used to select features for a given dataset based on a given algorithm. This method evaluates all possibilities against the evaluation criteria in a greedy search approach. Generally, wrapper methods lead to better predictive accuracy than filter methods.^{68–70,72} The most common techniques used in wrapper methods are Forward Feature Selection, Backward Feature Elimination, Exhaustive Feature Selection, and Recursive Feature Elimination.
- Embedded methods combine the advantages of both wrapper and filter methods while maintaining a reasonable level of interaction between features at a reasonable cost of computational complexity. The embedded methods are iterative, in that they take care of each iteration of the model training process and extract the features that are most important for that particular iteration.^{68–70,72} The most common techniques used in embedded methods are LASSO Regularization (L1), and Random Forest Importance.

In this systematic literature review, there were only 56 research articles that considered using feature selection methods/algorithms for their models to identify best features that are relevant for predicting students' performance. Forty nine (49) research articles explicitly mention their methods/algorithms for feature selection process, where 7 research articles

TABLE 4 List of feature selection methods/algorithms used in the literature.

Methods/algorithms	Frequency	Percent
Information Gain-based selection algorithms	16	23.5%
Correlation-based feature selection	7	10.0%
Gain Ratio Attribute Evaluator Filter (Gain Ratio)	6	9.0%
Relief Attribute Evaluation	5	7.5%
Chi Squared Attribute Evaluation (Chi)	4	6%
Principal Component Analysis (PCA)	3	4.4%
Binary Genetic Algorithm (BGA)	2	3.0%
Minimum Redundancy Maximum Relevance (mRMR)	2	3.0%
Pearson Correlation Coefficient	2	3.0%
Wrapper subset feature selection algorithms (WrapperSubsetEval)	2	3.0%
Alpha-investing feature selection method	1	1.5%
Boruta algorithm (a wrapper built around the Random Forest classification algorithm)	1	1.5%
C4.5	1	1.5%
F-regression	1	1.5%
F-test (filter-based method)	1	1.5%
Fast Correlation Based Filter (FCBF)	1	1.5%
Feature Selection based Sequential Minimal Optimization	1	1.5%
Genetic Algorithm (GA)	1	1.5%
Gini index	1	1.5%
Goodman and Kruskal's tau	1	1.5%
Harris Hawks Optimization (HHO) algorithm	1	1.5%
J48 Feature selection	1	1.5%
Laplacian Score	1	1.5%
Naive Bayes	1	1.5%
Nature-inspired feature algorithms	1	1.5%
One-R Attribute Evaluation	1	1.5%
Random Forest Regressor	1	1.5%
Scatter Search (SS)	1	1.5%
SelectKBest (Univariate feature selection available in sklearn)	1	1.5%
Symmetrical Uncertainty Attribute Evaluation	1	1.5%
Symmetrical Uncertainty Ranker	1	1.5%
Mean Absolute Difference (MAD)	1	1.5%
Total	68	100.0

did not indicate their methods. We have identified 68 distinct feature selection methods/algorithms used in the literature from the 49 research articles. As shown in Table 4, the top four methods/algorithms used in the literature are Information Gain-based selection algorithms, Correlation-based feature selection, Gain Ratio Attribute Evaluator Filter (Gain Ratio), and Relief Attribute Evaluation. Information Gain-based selection algorithms are the most algorithm used in the literature (23.5%). They have been used in 16 studies. Correlation-based feature selection was used in 7 research articles (10%), followed by Gain Ratio, which was used in six research articles (9%), and Relief Attribute Evaluation was used by 5 research articles (7.5%).



FIGURE 8 Most commonly used ML software and tools (percentages).

4.5 | RQ5: Student performance and learning environments

Next, we classify studies based on learning environments: e-learning, and traditional and blended learning environments (see Figure 7). Traditional and blended learning environments have face-to-face contact between instructors and students (e.g., on-campus labs, tutorials, etc.), while in e-learning, all learning interactions are online. Since all interactions in an e-learning environment are online, e-learning platforms have the ability to collect a more complete dataset of students' interactions and behavior using log files. Despite this, we observe that there are more studies in the literature (62% of the studies in this survey) using datasets from traditional and blended learning environments than e-learning (38%). These results address the RQ—*What learning environments have ML methods been applied to predict student performance*?

4.6 | RQ 6: Tools and platforms

This section outlines tools/platforms/libraries used to implement ML algorithms in the literature, addressing the RQ—"*What are the main ML tools/platforms/libraries used for predicting student performance?*". Although ML algorithms can be implemented from scratch, there are a number of programming language-specific libraries as well as tools/platforms available. It is no surprise that many studies in the literature used these libraries/tools/platforms to implement ML algorithms (Figure 8). It is noteworthy to mention that not all papers have specified the ML implementation language/tool used. Thus, Figure 8 is derived from 106 of the papers reviewed that have mentioned tools/platforms/libraries used in their studies. As seen in Figure 8, WEKA is the most common platform used in the literature (used by 42 studies, 40%) followed by Python (used by 34 studies, 32%).

4.7 | RQ 7: Remedial actions for at-risk students

Predicting student performance has received a lot of attention in the literature. However, few studies appear to have investigated the impacts of potential actions taken for detected at-risk students based on prediction results.^{64,73–78} In

these research studies, interventions have generally been determined by the instructor, who contacts the identified at-risk students personally (usually via email),⁷⁶ or who has pre-defined email templates that are automatically sent,^{75,77} or who manually contacts students anticipated to be at risk.⁶⁴ The effectiveness of these treatments has been demonstrated in lowering the dropout rate,⁶⁴ increasing retention,⁷⁶ enhancing students' achievements and grades,⁷⁵ and addressing the causes of why students struggled.⁷⁷ Although it is evident that actions based on predictions of student performance can have a substantial influence, we note that very few research studies have looked at this topic.

Purdue's course signals initiative was among the first to use student performance predictions in higher education.⁷⁶ Course signals' predictive models detect at-risk students using four data sources: current course grades, Blackboard engagement data, past academic history such high school GPAs, and demographic data such as residence and age. Based on the outcomes of the predictive models, the course signals system categorizes students into three groups: red signals indicate students are likely to fail, yellow signals suggest students might fail, and green signals indicate students are likely to pass their course. Academics create an intervention strategy based on the outcomes of the students' signals, which may consist of posting the students' signals on the LMS home pages, sending them customized emails or texts, referring them to academic advisors or academic resource centers, or scheduling a face-to-face meeting. Purdue University utilized course signals in 2007, 2008, and 2009 cohorts to assess the initiative's impact on student performance and retention.⁷⁶ Across several different cohorts at Purdue, the approach has increased both the success and retention rates.

Jayaprakash et al.⁷⁵ described the Open Academic Analytics Initiative, which creates an open-source platform to identify at-risk students early in the semester to enable interventions. Four data sources, student demographics, course grades and related data, student interaction data with the LMS, and students' progress on final grade so far, were used to determine whether a student was at risk of failing their courses. Logistic Regression, SVMs employing Sequential Minimal Optimization, J48 Decision Trees, and NB were initially used to develop predictive models. Logistic Regression performed better than other algorithms and was chosen as the eventual predictive model. Two intervention strategies-the "Awareness Messaging Intervention" and "Online Academic Support Environment Intervention"-were also evaluated in their study. The "Awareness Messaging Intervention" group received a message informing them that they were at risk of failing the course, and instructors were encouraged to provide actions (such as visits during office hours, set up appointments with tutors, access web-based resources, etc.). The "Online Academic Support Environment Intervention" group received a similar message to the "Awareness Intervention" group, except that instead of specific instructions, the students were invited to join the institution-wide online support services, such as access to Open Educational Resources instructional materials and a variety of mentoring from peers and professional support personnel (e.g., Khan Academy videos, Flat World Knowledge textbooks, etc.). In Spring 2012, a comparison between the control and intervention groups was performed (1739 students were enrolled in the study, with 451 student identified as being at risk). The findings indicated that there is a 6% improvement in the final grades in the two intervention groups compared to the non-intervention control group, and a statistically significant difference was achieved in final grades between the control group and intervention groups. There was no statistically significance difference between the two intervention groups.

Burgos et al.⁶⁴ were able to identify students who were likely to withdraw from a course early in the term based on their assessed activities. With 104 students across five courses, the prediction model was deployed to identify at-risk students who could otherwise drop out. Accuracy at week 10 (middle of the term) using the Logistic Regression model was 97.13%, precision at 98.95%, recall at 96.7%, and specificity at 97.1%. An intervention tutoring plan was also developed, allowing academics/instructors to intervene at different times during the semester (weeks 4, 7, and 10) to advise students at risk of dropping out of the course. The dropout rate was reduced by 14% compared with previous cohorts when instructors and academic advisors contacted them by email and phone.

A group of academics at a fully online university (Universitat Oberta de Catalunya (UOC)) have been working to create an Early Warning System (EWS) and an intervention strategy in order to identify students who are most likely to fail. The EWS employs a variety of classifiers, including NB, CART Decision Trees, K-NN, and SVMs.⁷⁷ The prediction models are created using data from the UOC data mart, which includes information on students' involvement and interaction (navigation, interaction, contact with LMSs, etc.), student profiles, and course performance data (student enrolment, assessment, etc.). The prediction results are shown to students using dashboards that show Green-Amber-Red signals. Additionally, the minimum marks necessary for the following assessment to receive a pass grade in the course are also shown.^{77,79,80} Interventions are automatically provided through email when at-risk students have been identified.⁸¹ The EWS and interventions are evaluated^{77,79–81} in a variety of courses at UOC. The results are encouraging, showing higher pass rates, lower dropout rates, and positive student and instructor acceptance. Bañeres et al.⁸² extended their work to predict students at risk of dropping out. When a student is detected as being at risk of dropping out a course, the intervention system automatically sends them personalized emails. The intervention was evaluated in a large online course at UOC and the dropout rate for students who participated in the intervention is significantly reduced (12% lower) than those who didn't at the end of the course.

Albreiki et al.⁸³ proposed a framework that employs ML algorithms and a rule-based model to identify students who are at risk of failing their courses and provide remedial actions. Student prediction models for two datasets using eight different ML algorithms to predict student performance as Good/At-Risk/Failed were developed. Dataset 1 is comprised of student performance data (also known as assessment data) for a course, and Dataset 2 is comprised of both student performance data and history data (such as student demographics, past performance, etc.). Out of the eight ML methods, ExtraTrees performed the best, with an accuracy score of 0.86 and an AUC score of 0.96 for Dataset 1, and 0.87 and 0.95, respectively, for Dataset 2. A framework for automatic remedial actions by mapping course learning outcomes with topics has been established, with the instructors determining a list of remedial actions that can be provided to predicted at-risk students at different times (checkpoints). The framework is yet to be implemented and tested in real-world settings.

Borrella et al.⁷⁴ evaluated four treatments to reduce dropouts in Massive Open Online Courses (MOOCs). The first two treatments used ML prediction models to identify and intervene with students at risk of dropping out. The first intervention consisted of an email sent to students at risk of dropping out with the intention of increasing their motivation before an important assessment. The second intervention provided students at risk of dropping out with exam preparation resources and study suggestions. In the third and fourth interventions, data analysis was used to identify at-risk students as intervention targets. In the third intervention, the difficulty of the assignments gradually increased. In the fourth intervention, the most difficult sections of the course were selected for redesign with scaffolding to help students, gradually enable students to improve understanding of the content. The first and second interventions had no statistically significant effect on the dropout rate. The third and fourth interventions revealed a statistically significant dropout rate reduction.

In conclusion, taking prompt actions based on identifying students who are at risk of failing and/or dropping out shows promise. However, it is evident that only a few studies have considered this important area, and this is a gap that future studies should consider exploring.

5 | DISCUSSION

This study reports a systematic literature review regarding the use of ML algorithms for predicting students' academic performance. The study aimed to identify:

- i. The aims of applying ML algorithms and methods in predicting student performance.
- ii. The ML methods and algorithms for predicting student performance;
- iii. The features/factors that affect the prediction of students' performance;
- iv. Methods employed for feature selection
- v. Learning environments where ML methods have been applied to predict student performance,
- vi. ML tools/platforms/libraries used to implement ML algorithms for predicting student performance, and
- vii. Actions taken based on the prediction results,

The study reviewed 162 research articles related to the subject of ML and specifically to EDM for predicting students' performance and came up with results of research distribution across multiple dimensions.

In terms of the first research direction, we have seen many studies that have applied ML algorithms, both supervised and unsupervised learning, to extract hidden knowledge and patterns from student data. Many applications have used classification (138 studies), regression (25 studies), or clustering (9 studies) methods in educational datasets to predict students' performance in their exams, predict if a student will be able to pass certain courses or not, predict students' CGPA, predict students' graduation time, and others. We classified broadly the main goals/aims of applying ML algorithms and methods for predicting student performance (in assessments, courses, or programs) as: (i) predicting student academic performance and identifying students at risk of failure, and (ii) predicting students at risk of dropping out. Our synthesis analysis showed that that most of the research studies have focused on predicting students' dropout and retention (33 studies). Interestingly, most of the dropout/retention prediction models are proposed in online or MOOC environments due to the highly prevalent issue of students dropping out in these environments. These findings are also consistent to findings in Albreiki et al.¹⁵

Features used for predicting student performance can vary based on several factors, such as data availability, the type of educational institution, specific courses, and goals of the prediction model. Our analysis revealed that historical records of students' grades and class performance were used in 122 studies, academic related features from LMSs were used in 98 studies, and students' demographic data was used in 97 studies. These are the top categories of features used for predicting students' performance. The findings here are consistent with those of Abu Saa et al.¹⁴ Shahiri and Husain,¹² and Albreiki et al.¹⁵, where historical records of students' grades and class performance, students' e-learning activity, and students' demographics are mainly considered for predicting students' performance. However, the relative importance of each feature will depend on the specific educational context and goals of the prediction model.

As aforementioned, three main approaches used for predicting students' performance in the literature are classification, regression, and clustering. More than 80% of the studies used classification to predict student performance. The main reasons is the nature of the problem where most research studies aimed to group/classify students who are at risk of either failing or dropping out the course. The second reason could be the nature of the dataset/feature used for creating prediction models, where most of the dataset are labelled and appropriate for supervised learning methods (i.e., classification or regression methods). Fewer studies (14.5%) have applied the regression approach, and only nine studies applied the unsupervised approach (i.e., clustering). Clustering is typically considered an exploratory technique because it helps discover patterns and groupings in data, rather than making predictions.

There are several reasons for supervised learning to be a popular approach for predicting student performance,^{1,84} including:

- i. Availability of labelled data: In supervised learning, the algorithm is trained on a labelled dataset, which means that the data includes the target variable (e.g., "pass" or "fail") and input features (e.g., past grades, attendance, demographics, etc.). This type of data is easily accessible in educational settings.
- ii. Predicting a continuous or categorical output: In the context of student performance prediction, the target variable is often a continuous value (e.g., a numerical grade) or a categorical value (e.g., pass/fail). Supervised learning algorithms, particularly regression and classification algorithms, are well suited for these types of prediction problems.
- iii. Understanding relationships between inputs and outputs: Supervised learning algorithms can uncover the relationships between input features and the target variable, which can provide valuable insights into what factors influence student performance.

However, the unsupervised approach (i.e., clustering) can still be used to make predictions about student performance by using the cluster assignments as features in a supervised learning model. It could also be used to improve the classification accuracy of predicting students' performance as in Almasri et al.,⁸⁵ where they proposed a cluster-based classifier that groups historical student records into a set of homogeneous clusters. By using clustering in this way, one can leverage the information gained from the exploratory analysis to make predictions about student performance. This approach can provide insights into the relationship between the features and the target variable, and can help improve the accuracy of the predictions.

There are many different ML algorithms used in the literature, more than 50 of them. Decision Tree, Random Forest, NB algorithms are the most commonly used.

We have found many different methods/algorithms in the literature for selecting features that are relevant to predictive models. These methods/algorithms are helpful, especially for large datasets, to reduce the amount of computational time and use of resources to develop a predictive ML model. However, we have seen less than 40% of research articles in the literature considered adopting feature selection methods/algorithms for their models. Many studies in the literature have tested their predictive models with and without using feature selection methods^{86–90}, and their models revealed better results when applying feature selection methods. Albreiki et al.'s systematic literature review¹⁵ also revealed that most of the research articles did not consider feature selection methods/algorithms for identifying features that are relevant to the prediction models.

There could be several reasons/factors why many research articles in the literature do not use feature selection methods in their student performance prediction models:

i. Insufficient data: if the data collected for the study is limited in size, feature selection may not be necessary, as the model will not over-fit the data and the number of features will be relatively low;

- ii. Data is already cleaned: if the data collected is already cleaned and does not contain a large number of redundant features;
- iii. Strong feature interpretation: if the researchers believe that the features included in the model are already highly informative and strongly interpretable; and
- iv. Computational complexity: feature selection methods can be computationally complex and time-consuming, and researchers may not have the resources or computational power to implement these methods.^{70,91–93}

These reasons/factors may influence a researcher's decision to use or not use feature selection methods in their studies. We have seen many research studies in the literature proposing models for predicting students' performance and identifying at-risk students. However, limited work exists on actions that can be taken based on these prediction results and evaluating their impact to meet educational outcomes. We found less than 10% of the research articles in this literature review that consider taking timely actions based on prediction results. These studies^{64,73–78} have shown promising results with improved student pass rates, reduced dropout rates, increases in retention, and enhanced students' achievements and grades.

As can be seen in this literature review, ML has great potential to predict students' performance and accurately identify students who are at risk. However, most of those predictive models are not easily interpretable/explainable to those, especially instructors, who are not familiar with them, which are often considered as black box models. When a model is used for decision-making, it is important to explain the reasons for a specific decision. Several benefits of using explainable ML models for predicting student performance include^{94–96}:

- Transparency: Explainable models provide a clear understanding of how the model is making its predictions, allowing stakeholders to understand the reasoning behind the predictions. This can help build trust in the model and ensure that its predictions are aligned with educational goals and values.
- Fairness: Explainable models can help identify potential sources of bias in the data used to train the model, allowing organizations to address these issues and ensure that the model is making fair predictions.
- Improved model performance: By understanding how the model is making predictions, stakeholders can identify areas where the model may be struggling and adjust the data or model to improve performance.
- Compliance: In some cases, regulations may require that organizations use explainable models, especially when sensitive data is involved.

Overall, the use of explainable ML models for predicting student performance can lead to better decision-making, improved model performance, and increased trust in the predictions made by the model. However, we found only two research articles^{83,97} in this literature review to have considered adopting the explainable ML predictive models. For example, Albreiki, et al.⁸³ proposed an explainable ML model in which academics are able to understand the causes affecting students' performance and why they are being classified as being at risk. Specifically, they provided visualizations for academics, on the 10 most important features that led to the prediction results, and demonstrated the strength of each feature in the prediction. These types of models are needed, especially when non–experts make decisions or take actions based on the predictive model results.

6 | CONCLUSION AND FUTURE RESEARCH DIRECTIONS

This literature review considered 162 papers published between January 2010 and October 2022, which aimed to predict student performance using ML methods. We classified the main goals of predicting student academic performance into two broad categories: (i) predicting student performance in assessments, courses or programs, and identifying students at risk of failure—129 studies; and (ii) predicting student dropout or retention in a course or program—33 studies. We also observed that classification is the most popular approach used in prediction of student performance (138 studies), followed by regression (25 studies) and clustering (9 studies). This makes sense as predicting student performance can be modelled as a classification problem, where students' results are grouped into different levels such as pass or fail as in binary classification or different grades in multiclass classification. Also, the historical data provides the dataset required for training such models. The supervised learning approach is widely used, as most datasets consist of labelled data. In terms of ML algorithms, more than 50 distinct ML algorithms were used by the 162 papers reviewed. The most commonly

-WILEY-Engineering Reports

used features for predicting students' performance are historical records of students' grades and class performance (presented in 122 studies), followed by academic related features from LMSs (presented in 98 studies) and students' demographics (presented in 97 studies). Around 30% of the research articles adopted feature selection methods for selecting features that are relevant to the prediction models. We identified distinct 32 feature selection methods/algorithms used in the literature. The most frequently used feature selection methods are Information Gain-based selection algorithm, Correlation-based feature selection, Gain Ratio, and Relief Attribute Evaluation. Most of the studies were conducted on data collected through traditional/blended learning environments compared to e-learning environments. We also identified 12 different tools/platforms/libraries that were used to implement ML algorithms for developing prediction models. Finally, very few research articles considered evaluating actions that can be taken and their impact based on the prediction results.

The EDM field aims to gain knowledge and insights from educational data for better outcomes in education. Through this study, we also observed areas that future research can focus on to result in better educational outcomes. One of the main goals of prediction is to identify at-risk students in the early stages and assist in them in real-time.^{54,59,60} However, in the literature review, we observed that most studies focus on prediction (data, models, accuracy, ML methods, etc.), but few focused on actions that can be taken based on the prediction results. Those studies that took actions based on prediction results demonstrated potential for better educational outcomes. Thus, we feel that there is potential to contribute in future as studies also focus on taking timely action based on prediction results. The challenge is to build systems that offer prediction of students' performance not only as an early and effective warning but also provide meaningful, valuable and timely feedback to students and interventions to improve outcomes.⁴⁸

Another area that can result in better outcomes in education from ML prediction results is identifying the causes/reasons for such prediction results. However, few studies focus on the cause of such results. In this literature review, we observed one study³⁹ that investigated the correlation between programs' admission requirements and criteria and students' performance. In their work, Adekitan and Noma-Osaghae³⁹ examined the association between Nigerian universities' acceptance requirements for choosing eligible prospective candidates for acceptance and the students' academic success during the first semester. Their results indicated that students' performance in the first year is not correlated to the entry requirements. In general, it is expected that there is a strong correlation of entry requirement to success in the programs to which they are used.

Investigating the correlation of learning activities to student performance can lead to better course design. García-Torres et al.⁹⁸ investigated and evaluated the relevance of Physics course activities to overall students' performance. Their results indicated that there are several activities that are not relevant or helpful towards students' performance in the course. These results can lead to new insights and improvements in course designs, activities, and assessments.

The course curriculum usually changes over years, and this process can impact the performance of the prediction models. We have seen many studies in the literature that use assessment data (students marks) as features in their prediction models. There is a possibility that in the next term/year, the weight of assessments is different, assessment types and the number of assessments can differ, and the prediction models' performance can be affected due to changes in assessments, weightage and others. This issue will probably affect the reliability of ML prediction models, especially in the first instance the course revisions are applied. Therefore, prediction models' reliability and accuracy need to consider changes in the data and updates to courses (and other features used in prediction models) as courses and programs evolve. Also, each iteration of the course provides additional data that can be used to further improve the accuracy of ML prediction models. Deep learning, a class of ML algorithms, may be better suited for prediction in evolving course structures and needs further investigation.

In summary, we hope that researchers can build on from the findings of this systematic literature review with an understanding of existing work on using ML to predict student performance, and also consider future research directions. While this study focused on how ML algorithms can predict student performance in general without considering features such as students' degree level, ethnicity, background, disabilities, university program and others, future literature reviews can expand the scope of the current work by considering how ML has been used in these different contexts. For instance, it would be interesting to look at what factors need to be considered in predicting the performance of international students, postgraduate coursework and research students, students with disabilities, indigenous students, students in different programs (Medicine vs. Computer Science vs. Humanities), and students from different countries, among others.

AUTHOR CONTRIBUTIONS

Khalid Alalawi: Data curation (lead); formal analysis (equal); investigation (equal); methodology (equal); validation (equal); writing – original draft (lead). **Rukshan Athauda:** Formal analysis (equal); investigation (equal); methodology (equal); project administration (equal); resources (equal); supervision (equal); validation (equal); writing – review and editing (equal). **Raymond Chiong:** Conceptualization (lead); investigation (equal); methodology (equal); project administration (equal); supervision (equal); validation (equal); methodology (equal); project administration (equal); resources (equal); supervision (equal); validation (equal); writing – review and editing (equal).

FUNDING INFORMATION

None.

CONFLICT OF INTEREST STATEMENT

The authors declare no potential conflict of interest

PEER REVIEW

The peer review history for this article is available at https://www.webofscience.com/api/gateway/wos/peer-review/10. 1002/eng2.12699.

DATA AVAILABILITY STATEMENT

Not applicable

ENDNOTE

*Complete College America (CCA) organization dashboard. Available online at https://completecollege.org/data-dashboard . Accessed on October 26th 2022.

ORCID

Raymond Chiong b https://orcid.org/0000-0002-8285-1903

REFERENCES

- 1. Imran M, Latif S, Mehmood D, Shah MS. Student academic performance prediction using supervised learning techniques. *Int J Emerg Technol Learn*. 2019;14(14):92-104. doi:10.3991/ijet.v14i14.10310
- 2. Algarni A. Data mining in education. Intl J Adv Comput Sci Appl. 2016;7(6):456-461.
- 3. Romero C, Ventura S. Data mining in education. WIREs. 2013;3(1):12-27. doi:10.1002/widm.1075
- 4. Mohamad SK, Tasir Z. Educational data mining: a review. Procedia Soc Behav Sci. 2013;97:320-324. doi:10.1016/j.sbspro.2013.10.240
- 5. Bishop CM. Pattern recognition and machine learning. Springer; 2006.
- 6. Budhi GS, Chiong R, Pranata I, Hu Z. Using machine learning to predict the sentiment of online reviews: a new framework for comparative analysis. *Arch Comput Methods Eng.* 2021;28:2543-2566. doi:10.1007/s11831-020-09464-8
- 7. Chiong R, Fan Z, Hu Z, Chiong F. Using an improved relative error support vector machine for body fat prediction. *Comput Methods Programs Biomed*. 2021;198:105749. doi:10.1016/j.cmpb.2020.105749
- Fischer C, Pardos ZA, Baker RS, et al. Mining big data in education: affordances and challenges. *Rev Res Educ.* 2020;44(1):130-160. doi:10.3102/0091732X20903304
- Livieris IE, Drakopoulou K, Kotsilieris T, Tampakas V, Pintelas P. DSS-PSP—A decision support software for evaluating students' performance. In: Iliadis L, Likas A, Jayne C, Boracchi G, eds. in 18th International Conference on Engineering Applications of Neural Networks, EANN 2017. Vol 744. Springer Verlag; 2017:63-74. doi:10.1007/978-3-319-65172-9_6
- 10. Roy S, Garg A. Analyzing performance of students by using data mining techniques a literature survey. 2017 4th IEEE Uttar Pradesh Section International Conference on Electrical, Computer and Electronics (UPCON). IEEE; 2017:130-133.
- 11. Christy AP, Rama N. Relevance and resonance of data science in performance prediction and visualization. *Indian J Sci Technol*. 2016;9(19):93885. doi:10.17485/ijst/2016/v9i19/93885
- 12. Shahiri AM, Husain W. A review on predicting student's performance using data mining techniques. Proc Comput Sci. 2015;72:414-422.
- 13. Kumar M, Singh A, Handa D. Literature survey on student's performance prediction in education using data mining techniques. *Int J Educ Manag Eng.* 2017;7(6):42-49.
- 14. Abu Saa A, Al-Emran M, Shaalan K. Factors affecting students' performance in higher education: a systematic review of predictive data mining techniques. *Technol Knowl Learn*. 2019;24(4):567-598. doi:10.1007/s10758-019-09408-7
- 15. Albreiki B, Zaki N, Alashwal H. A systematic literature review of student' performance prediction using machine learning techniques. *Educ Sci.* 2021;11(9):552. https://www.mdpi.com/2227-7102/11/9/552

^{22 of 25} WILEY - Engineering Reports

- ALALAWI ET AL
- Hellas A, Ihantola P, Petersen A, et al. Predicting academic performance: a systematic literature review. Presented at the Proceedings Companion of the 23rd Annual ACM Conference on Innovation and Technology in Computer Science Education, Larnaca, Cyprus. 2018. doi:10.1145/3293881.3295783
- 17. Manjarres AV, Sandoval LGM, Suárez MJS. Data mining techniques applied in educational environments: Literature review. *Dig Educ Rev.* 2018;33:235-266.
- Qazdar A, Er-Raha B, Cherkaoui C, Mammass D. A machine learning algorithm framework for predicting students performance: A case study of baccalaureate students in Morocco. *Educ Inf Technol.* 2019;24(6):3577-3589. doi:10.1007/s10639-019-09946-8
- 19. Chiong R, Hu Z, Fan Z, Lin Y, Chalup S, Desmet A. A bio-inspired clustering model for anomaly detection in the mining industry. *Bio-Inspired Computing Models and Algorithms.* World Scientific; 2019:133-155.
- 20. Lo SL, Chiong R, Cornforth D. An unsupervised multilingual approach for online social media topic identification. *Expert Syst Appl.* 2017;81:282-298. doi:10.1016/j.eswa.2017.03.029
- 21. Kostopoulos G, Kotsiantis S, Fazakis N, Koutsonikos G, Pierrakeas C. A Semi-Supervised Regression Algorithm for Grade Prediction of Students in Distance Learning Courses. *Int J Artif Intell Tools*. 2019;28(4):19. doi:10.1142/s0218213019400013
- 22. Ahmed ABED, Elaraby IS. Data mining: a prediction for student's performance using classification method. *World J Comput Appl Technol.* 2014;2(2):43-47.
- 23. Han J, Pei J, Kamber M. Data mining: concepts and techniques. Elsevier; 2011.
- 24. Madhulatha TS. An overview on clustering methods. arXiv preprint arXiv:1205.1117. 2012;2:719-725.
- 25. Zhu X, Goldberg AB. Introduction to semi-supervised learning. Synth Lect Artif Intell Mach Learn. 2009;3(1):1-130.
- Lo SL, Cambria E, Chiong R, Cornforth D. A multilingual semi-supervised approach in deriving Singlish sentic patterns for polarity detection. *Knowl Based Syst.* 2016;105:236-247. doi:10.1016/j.knosys.2016.04.024
- 27. Zhou Y, Goldman S. Democratic co-learning. 16th IEEE International Conference on Tools with Artificial Intelligence, 15–17. IEEE; 2004:594-602. doi:10.1109/ICTAI.2004.48
- 28. Kitchenham B, Pretorius R, Budgen D, et al. Systematic literature reviews in software engineering-a tertiary study. *Inform Softw Technol*. 2010;52(8):792-805.
- 29. Kotsiantis S, Patriarcheas K, Xenos M. A combinational incremental ensemble of classifiers as a technique for predicting students' performance in distance education. *Knowl Based Syst.* 2010;23(6):529-535. doi:10.1016/j.knosys.2010.03.010
- 30. Kotsiantis SB. Use of machine learning techniques for educational proposes: A decision support system for forecasting students' grades. *Artif Intell Rev.* 2012;37(4):331-344. doi:10.1007/s10462-011-9234-x
- Thai-Nghe N, Drumond L, Krohn-Grimberghe A, Schmidt-Thieme L. Recommender system for predicting student performance. Proc Comput Sci. 2010;1(2):2811-2819. doi:10.1016/j.procs.2010.08.006
- 32. Ahadi A, Lister R, Haapala H, Vihavainen A. Exploring machine learning methods to automatically identify students in need of assistance. 11th Annual ACM Conference on International Computing Education Research, ICER 2015. Association for Computing Machinery, Inc, 121–130. 2015 https://www.scopus.com/inward/record.uri?eid=2-s2.0-84959308908&doi=10.1145%2f2787622.2787717&partnerID=40& md5=e102310811df2e532d8ed2f4839fea80
- Hussain M, Zhu W, Zhang W, Abidi SMR, Ali S. Using machine learning to predict student difficulties from learning session data. Artif Intell Rev. 2019;52(1):381-407. doi:10.1007/s10462-018-9620-8
- 34. Guo B, Zhang R, Xu G, Shi CM, Yang L. Predicting students performance in educational data mining. In: Wang FL, Wong TL, Au O, Liu QT, Wu D, eds. 2015 International Symposium on Educational Technology. IEEE; 2015:125-128. doi:10.1109/iset.2015.33
- 35. Adekitan AI, Salau O. The impact of engineering students' performance in the first three years on their graduation result using educational data mining. *Heliyon*. 2019;5(2):e01250. doi:10.1016/j.heliyon.2019.e01250
- 36. Buenaño-Fernández D, Gil D, Luján-Mora S. Application of machine learning in predicting performance for computer engineering students: a case study. *Sustainability*. 2019;11(10):2833. doi:10.3390/su11102833
- Wang L, Sy A, Liu L, Piech C. Deep knowledge tracing on programming exercises. Proceedings of the Fourth (2017) ACM Conference on Learning @ Scale. ACM; 2017:201-204. https://www.scopus.com/inward/record.uri?eid=2-s2.0-85018422818&doi=10.1145%2f3051457. 3053985&partnerID=40&md5=0aca95a882db49611a9edabebdbab468
- Ghorbani R, Ghousi R. Comparing different resampling methods in predicting students' performance using machine learning techniques (in English). *IEEE Access*. 2020;8:67899-67911. doi:10.1109/ACCESS.2020.2986809
- Adekitan AI, Noma-Osaghae E. Data mining approach to predicting the performance of first year student in a university using the admission requirements. *Educ Inf Technol.* 2019;24(2):1527-1543. doi:10.1007/s10639-018-9839-7
- 40. Yoo J, Kim J. Can online discussion participation predict group project performance? Investigating the roles of linguistic features and participation patterns. *Int J Artif Intell Educ.* 2014;24(1):8-32. doi:10.1007/s40593-013-0010-8
- Injadat M, Moubayed A, Nassif AB, Shami A. Systematic ensemble model selection approach for educational data mining (in English). Knowl Based Syst. 2020;200:16. doi:10.1016/j.knosys.2020.105992
- 42. Wang L, Sy A, Liu L, Piech C. Learning to represent student knowledge on programming exercises using deep learning. In: Hu X, Hu X, Barnes T, Hershkovitz A, Paquette L, eds. 10th International Conference on Educational Data Mining, EDM 2017. International Educational Data Mining Society; 2017:324-329. https://www.scopus.com/inward/record.uri?eid=2-s2.0-85071328238&partnerID=40&md5=9fec4f770f8f69ac5710f1e15047cfea
- Tsiakmaki M, Kostopoulos G, Kotsiantis S, Ragos O. Transfer learning from deep neural networks for predicting student performance (in English). *Appl Sci Basel*. 2020;10(6):12. doi:10.3390/app10062145

- 44. Jalota C, Agrawal R. Analysis of educational data mining using classification. Proceedings of the International Conference on Machine Learning, Big Data, Cloud and Parallel Computing: Trends, Prespectives and Prospects, COMITCon 2019, 243–247. 2019 https://www.scopus.com/inward/record.uri?eid=2-s2.0-85074164419&doi=10.1109%2fCOMITCon.2019.8862214&partnerID=40& md5=df6d0d850f45f62a6fd8a9c6ae8b1a42
- 45. Livieris IE, Drakopoulou K, Tampakas VT, Mikropoulos TA, Pintelas P. Predicting secondary school students' performance utilizing a semi-supervised learning approach. *J Educ Comput Res.* 2019;57(2):448-470. doi:10.1177/0735633117752614
- 46. Almasri A, Celebi E, Alkhawaldeh RS. EMT: ensemble meta-based tree model for predicting student performance. *Sci Program.* 2019;2019:13. doi:10.1155/2019/3610248
- 47. Khan I, Al Sadiri A, Ahmad AR, Jabeur N. Tracking student performance in introductory programming by means of machine learning. 4th MEC International Conference on Big Data and Smart City, ICBDSC 2019, 15–16 Jan. 2019, IEEE, 1–6. 2019 https://www.scopus.com/inward/record.uri?eid=2-s2.0-85063188659&doi=10.1109%2fICBDSC.2019.8645608&partnerID=40&md5= bdaddfee7e4aceae0c72beb606b95d7c
- 48. Cano A, Leonard JD. Interpretable multiview early warning system adapted to underrepresented student populations. *IEEE Trans Learn Technol.* 2019;12(2):198-211. doi:10.1109/tlt.2019.2911079
- 49. Beemer J, Spoon K, He LJ, Fan JJ, Levine RA. Ensemble learning for estimating individualized treatment effects in student success studies. Int J Artif Intell Educ. 2018;28(3):315-335. doi:10.1007/s40593-017-0148-x
- 50. Tsiakmaki M, Kostopoulos G, Kotsiantis S, Ragos O. Implementing autoML in educational data mining for prediction tasks. *Appl Sci Basel*. 2020;10(1):90. doi:10.3390/app10010090
- Injadat MN, Moubayed A, Nassif AB, Shami A. Multi-split optimized bagging ensemble model selection for multi-class educational data mining (in English). *Appl Intell*. 2020;50(12):4506-4528. doi:10.1007/s10489-020-01776-3
- 52. Hashim AS, Awadh WA, Hamoud AK. Student performance prediction model based on supervised machine learning algorithms. In: Shafik SS, Roomi AB, Sharrad FI, eds. 2nd International Scientific Conference of Al-Ayen University, ISCAU 2020. Vol 928. 3rd ed. IOP Publishing Ltd; 2020. https://www.scopus.com/inward/record.uri?eid=2-s2.0-85097135865&doi=10.1088%2f1757-899X%2f928%2f3%2f032019&partnerID=40&md5=eeb05de132771f68a93a7c11f65e8303
- Crivei LM, Ionescu VS, Czibula G. An analysis of supervised learning methods for predicting students' performance in academic environments. *ICIC Express Lett.* 2019;13(3):181-189. doi:10.24507/icicel.13.03.181
- 54. Kovacic Z. Early prediction of student success: Mining students' enrolment data. *Informing Science + Information Technology Education Joint Conference*, Cassino, Italy. 2010.
- 55. Kostopoulos G, Karlos S, Kotsiantis S. Multiview learning for early prognosis of academic performance: a case study. *IEEE Trans Learn Technol.* 2019;12(2):212-224. doi:10.1109/TLT.2019.2911581
- Athani SS, Kodli SA, Banavasi MN, Hiremath PGS. Student performance predictor using multiclass support vector classification algorithm. Proceedings of IEEE International Conference on Signal Processing and Communication, ICSPC 2017, vol. 2018, 341–346. 2018.
- 57. Chiong R, Fan Z, Hu Z, Dhakal S. A novel ensemble learning approach for stock market prediction based on sentiment analysis and the sliding window method. *IEEE Transactions on Computational Social Systems*. 2023. doi:10.1109/TCSS.2022.3182375
- 58. Chiong R, Wang Z, Fan Z, Dhakal S. A fuzzy-based ensemble model for improving malicious web domain identification. *Expert Systems with Applications*. 2022;204:117243.
- 59. Hu Y-H, Lo C-L, Shih S-P. Developing early warning systems to predict students' online learning performance. *Comput Hum Behav*. 2014;36:469-478.
- Hung J, Wang MC, Wang S, Abdelrasoul M, Li Y, He W. Identifying at-risk students for early interventions—a time-series clustering approach. *IEEE Trans Emerg Top Comput.* 2017;5(1):45-55. doi:10.1109/TETC.2015.2504239
- Trakunphutthirak R, Cheung Y, Lee V. Detecting student at risk of failure: A case study of conceptualizing mining from internet access log files. In: Yu J, Tong H, Zhu F, Li Z, eds. 18th IEEE International Conference on Data Mining Workshops, ICDMW 2018. Vol 2018. IEEE Computer Society; 2019:365-371. https://www.scopus.com/inward/record.uri?eid=2-s2.0-85062822011&doi=10.1109%2fICDMW. 2018.00060&partnerID=40&md5=0baa6b14dc921a1d3cde7f6fde6c30e3
- 62. Iqbal Z, Qayyum A, Latif S, Qadir J. Early student grade prediction: an empirical study. *in 2019 2nd International Conference on Advancements in Computational Sciences (ICACS)*. IEEE; 2019:1-7. doi:10.23919/ICACS.2019.8689136
- 63. Livieris IE, Tampakas V, Karacapilidis N, Pintelas P. A semi-supervised self-trained two-level algorithm for forecasting students' graduation time. *Intell Decis Technol.* 2019;13(3):367-378. doi:10.3233/IDT-180136
- 64. Burgos C, Campanario ML, de la Pena D, Lara JA, Lizcano D, Martínez MA. Data mining for modeling students' performance: A tutoring action plan to prevent academic dropout. *Comput Electr Eng.* 2018;66:541-556.
- 65. Iam-On N, Boongoen T. Improved student dropout prediction in Thai University using ensemble of mixed-type data clusterings. *Int J Mach Learn Cybern.* 2017;8(2):497-510.
- 66. Aulck L, Velagapudi N, Blumenstock J, West J. Predicting student dropout in higher education, *arXiv preprint arXiv:1606.06364*. 2016.
- 67. Lee S, Chung JY. The Machine Learning-Based Dropout Early Warning System for Improving the Performance of Dropout Prediction. *Appl Sci.* 2019;9(15):3093.
- Jović A, Brkić K, Bogunović N. A review of feature selection methods with applications. 2015 38th International Convention on Information and Communication Technology, Electronics and Microelectronics (MIPRO), 25–29 May, 1200–1205. 2015. doi:10.1109/MIPRO.2015.7160458
- 69. Remeseiro B, Bolon-Canedo V. A review of feature selection methods in medical applications. *Comput Biol Med.* 2019;112:103375. doi:10.1016/j.compbiomed.2019.103375

WILEY-Engineering Reports

- 70. Bolón-Canedo V, Sánchez-Maroño N, Alonso-Betanzos A. A review of feature selection methods on synthetic data. *Knowl inform Syst.* 2013;34:483-519.
- 71. Guyon I, Gunn S, Nikravesh M, Zadeh LA. Feature extraction: foundations and applications. Springer; 2008.
- 72. Ezz M, Elshenawy A. Adaptive recommendation system using machine learning algorithms for predicting student's best academic program. *Educ Inf Technol.* 2019;25:14-2746. doi:10.1007/s10639-019-10049-7
- 73. Bañeres D, Rodríguez-González ME, Guerrero-Roldán A-E, Cortadas P. An early warning system to identify and intervene online dropout learners. *Int J Educ Technol High Educ.* 2023;20(1):3-25. doi:10.1186/s41239-022-00371-5
- 74. Borrella I, Caballero-Caballero S, Ponce-Cueto E. Taking action to reduce dropout in MOOCs: Tested interventions. *Comput Educ.* 2022;179:104412. doi:10.1016/j.compedu.2021.104412
- 75. Jayaprakash SM, Moody EW, Lauría EJM, Regan JR, Baron JD. Early alert of academically at-risk students: an open source analytics initiative. *J Learn Anal*. 2014;1(1):6-47. doi:10.18608/jla.2014.11.3
- 76. Arnold KE, Pistilli MD. Course signals at Purdue: using learning analytics to increase student success. Presented at the Proceedings of the 2nd International Conference on Learning Analytics and Knowledge, Vancouver, British Columbia, Canada. 2012. doi:10.1145/2330601.2330666
- 77. Baneres D, Rodríguez-Gonzalez ME, Serra M. An early feedback prediction system for learners at-risk within a first-year higher education course. *IEEE Trans Learn Technol.* 2019;12(2):249-263. doi:10.1109/TLT.2019.2912167
- 78. Figueroa-Cañas J, Sancho-Vinuesa T. Changing the recent past to reduce ongoing dropout: an early learning analytics intervention for an online statistics course. *Open Learn*. 2021;36:1-18. doi:10.1080/02680513.2021.1971963
- 79. Bañeres D, Rodríguez ME, Guerrero-Roldán AE, Karadeniz A. An early warning system to detect at-risk students in online higher education. *Appl Sci.* 2020;10(13):4427.
- 80. Guerrero-Roldán A-E, Rodríguez-González ME, Bañeres D, Elasri-Ejjaberi A, Cortadas P. Experiences in the use of an adaptive intelligent system to enhance online learners' performance: a case study in Economics and Business courses. *Int J Educ Technol High Educ.* 2021;18:1-27.
- 81. Rodríguez ME, Guerrero-Roldán AE, Baneres D, Karadeniz A. An intelligent nudging system to guide online learners. *Int Rev Res Open Distance Learn*. 2022;23(1):41-62.
- 82. Bañeres D, Rodríguez-González ME, Guerrero-Roldán A-E, Cortadas P. An early warning system to identify and intervene online dropout learners. *Int J Educ Technol High Educ.* 2023;20(1):1-25.
- 83. Albreiki B, Habuza T, Zaki N. Framework for automatically suggesting remedial actions to help students at risk based on explainable ML and rule-based models. *Int J Educ Technol High Educ*. 2022;19(1):49-26. doi:10.1186/s41239-022-00354-6
- 84. Cunningham P, Cord M, Delany SJ. Supervised learning. In: Cord M, Cunningham P, eds. *Machine Learning Techniques for Multimedia: Case Studies on Organization and Retrieval*. Springer Berlin Heidelberg; 2008:21-49.
- Almasri A, Alkhawaldeh RS, Celebi E. Clustering-based EMT model for predicting student performance (in English). Arab J Sci Eng. 2020; 45:12. doi:10.1007/s13369-020-04578-4
- 86. García-Torres M, Becerra-Alonso D, Gómez-Vela FA, Divina F, Cobo IL, Martínez-Álvarez F. Analysis of student achievement scores: a machine learning approach. Presented at the International Joint Conference: 12th International Conference on Computational Intelligence in Security for Information Systems, CISIS 2019 and 10th International Conference on European Transnational Education, ICEUTE 2019, Conference Paper. 2020.
- 87. Turabieh H. Hybrid machine learning classifiers to predict student performance. 2019 2nd International Conference on new Trends in Computing Sciences (ICTCS). IEEE; 2019:1-6.
- Almasri A, Alkhawaldeh RS, Çelebi E. Clustering-Based EMT Model for Predicting Student Performance (in English). Arab J Sci Eng. 2020;45(12):10067-10078. doi:10.1007/s13369-020-04578-4
- 89. Ismail L, Materwala H, Hennebelle A. Comparative Analysis of Machine Learning Models for Students' Performance Prediction. *Adv Intell Syst Comput.* 2021;1352:149-160.
- 90. Dake DK, Essel DD, Agbodaze JE. Using machine learning to predict students' academic performance during COVID-19. Proceedings—2021 International Conference on Computing, Computational Modelling and Applications, ICCMA 2021, 9–15. 2021 https://www.scopus.com/inward/record.uri?eid=2-s2.0-85126022284&doi=10.1109%2fICCMA53594.2021.00010&partnerID=40& md5=640f1b9552ec0803b937a563756ea607
- 91. Hira ZM, Gillies DF. A review of feature selection and feature extraction methods applied on microarray data. *Adv Bioinform*. 2015;2015:1-13.
- 92. Saeys Y, Inza I, Larranaga P. A review of feature selection techniques in bioinformatics. *Bioinformatics*. 2007;23 (19):2507-2517.
- Khalid S, Khalil T, Nasreen S. A survey of feature selection and feature extraction techniques in machine learning. 2014 Science and Information Conference, 27–29 Aug, 372–378. 2014. doi:10.1109/SAI.2014.6918213
- 94. Guleria P, Sood M. Explainable AI and machine learning: performance evaluation and explainability of classifiers on educational data mining inspired career counseling. *Educ Inf Technol.* 2022;28:1-36.
- 95. Ribeiro MT, Singh S, Guestrin C. Why should i trust you? Explaining the predictions of any classifier. *Proceedings of the 22nd ACM SIGKDD International Conference on Knowledge Discovery and Data Mining*, 1135–1144. 2016.
- 96. Swamy V, Radmehr B, Krco N, Marras M, Käser T. Evaluating the explainers: black-box explainable machine learning for student success prediction in MOOCs. *arXiv preprint arXiv:2207.00551.* 2022.

- 97. Alharbi B. Back to basics: an interpretable multi-class grade prediction framework. *Arab J Sci Eng.* 2022;47(2):2171-2186. doi:10.1007/s13369-021-06153-x
- 98. García-Torres M, Becerra-Alonso D, Gómez-Vela FA, Divina F, Cobo IL, Martínez-Álvarez F. Analysis of student achievement scores: a machine learning approach. In: Martinez Alvarez F, Troncoso Lora A, Saez Munoz JA, Corchado E, Quintian H, eds. International Joint Conference: 12th International Conference on Computational Intelligence in Security for Information Systems, CISIS 2019 and 10th International Conference on European Transnational Education, ICEUTE 2019. Vol 951. Springer Verlag; 2020:275-284. doi:10.1007/978-3-030-20005-3_28

SUPPORTING INFORMATION

Additional supporting information can be found online in the Supporting Information section at the end of this article.

How to cite this article: Alalawi K, Athauda R, Chiong R. Contextualizing the current state of research on the use of machine learning for student performance prediction: A systematic literature review. *Engineering Reports*. 2023;5(12):e12699. doi: 10.1002/eng2.12699